Author's response to reviews

Title: Gross Haematuria From a Routine Run: An unusual presentation of congenital renal arteriovenous malformation.

Authors:

   Erin L Dames (erinleslee@gmail.com)
   Lay Guat Ng (ng.lay.guat@sgh.com.sg)
   Kiang Hiong Tay (tay.kiang.hiong@sgh.com.sg)

Version: 2 Date: 19 September 2013

Author's response to reviews:

To whom it may concern:

After revision and reformatting according to the Journal of Medical Case Reports stipulations, I, Erin Leslee Dames, resubmit a case report with the revised title: "Congenital Renal Arteriovenous Malformation presenting with gross haematuria after a routine jog: a case report." This case is presented due to its unusual presentation for a congenital renovascular anomaly in order to highlight its importance in the diagnosis of acute upper urinary tract haemorrhage.